Author's response to reviews

Title: Perception versus polysomnographic assessment of sleep in CFS and non-fatigued control subjects: results from a population-based study

Authors:

Matthias Majer (mmajer@emory.edu)
James F Jones (jaj9@cdc.gov)
Elizabeth R Unger ( eru0@cdc.gov)
Laura Solomon Youngblood (zfk9@cdc.gov)
Michael J Decker (mdecker@fusionsleep.com)
Brian Gurbaxani (buw8@cdc.gov)
Christine Heim (cmheim@emory.edu)
William C Reeves (wcr1@cdc.gov)

Version: 5 Date: 3 December 2007

Author's response to reviews:

The editorial team caught one more stupid error on my part - two references #26. I fixed the dual numbers and subsequent numbers